Author's response to reviews

**Title:** Bilateral ovarian teratoma complicated with carcinosarcoma in a 68 year old woman: a case report

**Authors:**

Santhanam Shanmughapriya (shanmughapriyasanthanam@gmail.com)
Ganesan Senthilkumar (oncosphynx@yahoo.co.in)
Kumbalingam Balakrishnan (dr.kb62@gmail.com)
Nachiappan Vasanthi (vasanthinr@gmail.com)
Krishnakumar Vinodhini (vinubiotech25@yahoo.co.in)
Kalimuthusamy Natarajaseenivasan (natarajaseenivasan@rediffmail.com)

**Version:** 4 **Date:** 22 April 2011

**Author's response to reviews:** see over
Dear Editor,

The manuscript entitled “Bilateral ovarian teratoma complicated with carcinosarcoma in a 68 year old woman: a case report” has been edited extensively as per the comments of the reviewers. Herewith I am submitting the revised manuscript along with the point by point response to the editor’s comments for your kind perusal and publication in your esteemed journal.

Thanks & Regards

K. Natarajaseenivasan

Associate Editor

<table>
<thead>
<tr>
<th>No</th>
<th>Editor’s Comments</th>
<th>Author’s respond</th>
</tr>
</thead>
<tbody>
<tr>
<td>1</td>
<td>The authors just need to use the term immature teratoma in the body of the text instead of mature teratoma</td>
<td>The term mature teratoma has been changed to immature teratoma in the body of the text and necessary supporting references given</td>
</tr>
</tbody>
</table>